\documentclass[12pt,a4paper,nocenter,bold]{thema}%,twoside

\usepackage [dvips] {graphicx}
\usepackage{ams}
\usepackage{appendix}

%\setlength\paperheight{297mm}
%\setlength\paperwidth{210mm}
%\newlength{\lmargin}
%\newlength{\rmargin}
%\setlength{\lmargin}{2cm} \setlength{\rmargin}{0.5cm}
%%\setlength{\lmargin}{2-4cm} \setlength{\rmargin}{0.5cm}
%\setlength{\textwidth}{\paperwidth} 
%\setlength{\hoffset}{-4cm}

%\addtolength{\hoffset}{\lmargin}
%\addtolength{\textwidth}{-\lmargin}
%\addtolength{\textwidth}{-\rmargin}
\setlength{\textheight}{22cm}
%\setlength{\textwidth}{22cm}


%\addtolength{\evensidemargin}{-2.5cm}
%\addtolength{\oddsidemargin}{-0.5cm}

\addtolength{\textwidth}{1.5cm}

\begin{document}
\thispagestyle{empty}
\begin{titlepage}
\begin{center}
\begin{LARGE}Genome Sequencing: Algorithms for New Approaches\end{LARGE}
\vskip 0.5cm
Sekwencjonowanie genomu: algorytmy dla nowych podej\'{s}\'{c}
\vskip 0.5cm
\vskip 1.5cm
\begin{Large}Aleksandra \'{S}wiercz\end{Large}
\vskip 4cm
\vskip 0.5cm
Ph.D. Thesis
\vskip 0.5cm
\vskip 0.5cm
\vskip 0.5cm
%INFORMATYKI
\vskip 2cm
\textit{Promotor: dr hab. in\.z. Marta Kasprzak}
\vskip 2cm
Poznan University of Technology
\vskip 0.5cm
Institute of Computing Science
\vskip 1.4cm
Pozna\'{n}, 2007
\end{center}
%\cleardoublepage
\end{titlepage}
%\thispagestyle{empty}

%------------ body of article ------------------->>
\chapter{Introduction}
\section{Genome sequencing and the role of bioinformatics}

Published in \cite{ronaghi96,aarts97,bains88} 

Look here\cite{blazewicz88,cormen94, chen06}
or here \cite{fogel98,hgp03,patent95}


In Chapter \ref{algorithms} 

\chapter{New algorithms for DNA sequencing by hybridization with isothermic libraries}\label{algorithms}

\bibliographystyle{alpha}
\addcontentsline{toc}{chapter}{Bibliography}
\bibliography{baza}
%\nocite{*}

\end{document}

